Author's response to reviews

Title: The relative efficacy of nine osteoporosis medications for reducing the rate of fractures in post-menopausal women

Authors:

Robert B Hopkins (hopkinr@mcmaster.ca)
Ron Goeree (goereer@mcmaster.ca)
Eleanor Pullenayegum (pullena@mcmaster.ca)
Jonathan D Adachi (jd.adachi@sympatico.ca)
Alexandra Papaioannou (PAPAIOANNOU@HHSC.CA)
Feng Xie (fengxie@mcmaster.ca)
Lehana Thabane (thabanl@mcmaster.ca)

Version: 5 Date: 23 September 2011

Author's response to reviews:

Dear Editors

I have addressed the requested revisions.
The white spaces and titles have been removed from the figures.
The title and legends for the Figures and Additional Files are listed at the end of the text.
Vertical lines have been removed from the tables.

Sincerely
Robert Hopkins